UC San Diego UC San Diego Previously Published Works

Title

Antineoplastic kinase inhibitors: A new class of potent anti-amoebic compounds

Permalink https://escholarship.org/uc/item/0tz4754w

Journal PLOS Neglected Tropical Diseases, 15(2)

ISSN 1935-2727

Authors

Sauvey, Conall Ehrenkaufer, Gretchen Shi, Da <u>et al.</u>

Publication Date

2021

DOI

10.1371/journal.pntd.0008425

Peer reviewed



G OPEN ACCESS

Citation: Sauvey C, Ehrenkaufer G, Shi D, Debnath A, Abagyan R (2021) Antineoplastic kinase inhibitors: A new class of potent anti-amoebic compounds. PLoS Negl Trop Dis 15(2): e0008425. https://doi.org/10.1371/journal.pntd.0008425

Editor: Kiyoshi Kita, University of Tokyo, JAPAN

Received: May 21, 2020

Accepted: December 21, 2020

Published: February 8, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0008425

Copyright: © 2021 Sauvey et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

Funding: AD received funding from the United States National Institutes of Health (grant no. R21Al146460). CS received funding from the

RESEARCH ARTICLE

Antineoplastic kinase inhibitors: A new class of potent anti-amoebic compounds

Conall Sauvey^{1*}, Gretchen Ehrenkaufer², Da Shi¹, Anjan Debnath¹, Ruben Abagyan^{1*}

 Center for Discovery and Innovation in Parasitic Diseases, Skaggs School for Pharmacy and Pharmaceutical Sciences, University of California—San Diego, La Jolla, California, United States of America,
Division of Infectious Diseases, Department of Internal Medicine, Stanford University School of Medicine, Stanford, California, United States of America

* csauvey@gmail.com (CS); rabagyan@health.ucsd.edu (RA)

Abstract

Entamoeba histolytica is a protozoan parasite which infects approximately 50 million people worldwide, resulting in an estimated 70,000 deaths every year. Since the 1960s E. histolytica infection has been successfully treated with metronidazole. However, drawbacks to metronidazole therapy exist, including adverse effects, a long treatment course, and the need for an additional drug to prevent cyst-mediated transmission. E. histolytica possesses a kinome with approximately 300-400 members, some of which have been previously studied as potential targets for the development of amoebicidal drug candidates. However, while these efforts have uncovered novel potent inhibitors of E. histolytica kinases, none have resulted in approved drugs. In this study we took the alternative approach of testing a set of twelve previously FDA-approved antineoplastic kinase inhibitors against E. histolytica trophozoites in vitro. This resulted in the identification of dasatinib, bosutinib, and ibrutinib as amoebicidal agents at low-micromolar concentrations. Next, we utilized a recently developed computational tool to identify twelve additional drugs with human protein target profiles similar to the three initial hits. Testing of these additional twelve drugs led to the identification of ponatinib, neratinib, and olmutinib were identified as highly potent, with EC₅₀ values in the sub-micromolar range. All of these six drugs were found to kill E. histolytica trophozoites as rapidly as metronidazole. Furthermore, ibrutinib was found to kill the transmissible cyst stage of the model organism E. invadens. Ibrutinib thus possesses both amoebicidal and cysticidal properties, in contrast to all drugs used in the current therapeutic strategy. These findings together reveal antineoplastic kinase inhibitors as a highly promising class of potent drugs against this widespread and devastating disease.

Author summary

Every year, nearly a hundred thousand people worldwide die from infection by the intestinal parasite *Entamoeba histolytica*, despite the widespread availability of metronidazole as a treatment. Here we report that six anticancer drugs of the kinase inhibitor class possess potent anti-amoebic properties, with one of them killing both actively dividing parasite University of California - San Diego Global Health Institute (http://globalhealth.ucsd.edu/grantrecipients/2017- 2018/Pages/

GraduateStudentResearchers.aspx) (GHI Graduate Student research grant 2017-2018) GE received funding from the SPARK Translational Research Program at Stanford University (http://med. stanford.edu/sparkmed.html), the United States National Institute of Health (https://www.nih.gov/) (grant no. R21-AI123594), and the United States National Institute of Health National Center for Advancing Translational Science (https://ncats.nih. gov/) (grant no. UL1-TR001085) DS and AD received funding from the United States National Institute of Health (https://www.nih.gov/) (grant no. 1KL2TR001444) RA received funding from the United States National Institute of Health National Institute of General Medical Sciences (https://www. nigms.nih.gov/) (grant no. GM071872) The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

and its transmissible cysts. These anticancer kinase inhibitors, including the dual-purpose drug with both amoebicidal and cysticidal activities may be used to treat amoebiasis, especially in cancer patients or in life-threatening brain- and liver-infecting forms of the disease.

Introduction

Entamoeba histolytica is a parasitic amoeba which infects an estimated 50 million people worldwide, resulting in around 70,000 deaths per year [1]. *E. histolytica* infection is known as amoebiasis and primarily affects the intestinal tract in humans, most commonly causing symptoms such as abdominal pain, bloody diarrhea, and colitis [2]. In rare cases the infection spreads to other organs such as the liver and brain, and in serious cases results in patient death [2]. *E. histolytica*'s life cycle consists of a trophozoite vegetative stage which matures in its host to an infective cyst stage. The cyst stage is excreted in the host's feces, infecting a new host when ingested via a route such as drinking contaminated water. In the majority of cases where *E. histolytica* is ingested it lives asymptomatically in the human host's intestinal tract. Symptoms can develop when compromise of the mucosal layer allows it to come into contact with the intestinal wall, at which point it invades the wall and surrounding tissue causing characteristic 'flask-shaped ulcers' [3]. Due to this mode of transmission *E. histolytica* disproportion-ately affects populations experiencing sanitation problems associated with low socioeconomic status [2,4,5]. Malnutrition is also known to be a major risk factor for amoebiasis, especially in children [6].

E. histolytica infection is currently treated with the 5-nitroimidazole drug metronidazole, which has been in use since the 1960s and has widespread use as a treatment against anaerobic microbial infection [7,8]. However, while successful, metronidazole is not a perfect solution to *E. histolytica* infection, with a few particularly notable existing issues. One of these is problems with lack of patient compliance with the full course of treatment, leading to relapses and increased disease spread [7]. This is possibly due to factors such as drug adverse effects or the need for continued dosing past the resolution of disease symptoms [9,10]. Another issue is metronidazole's inability to kill the infective cyst stage of *E. histolytica*. Because of this, along with its complete absorbance from the intestines, metronidazole must be followed by a secondary luminal amoebicide such as paromomycin to prevent spread of the disease [11,12]. Also concerning is the potential for the emergence of resistance to metronidazole, which has been previously observed in the laboratory [13]. When considered together, these factors comprise an unmet need for alternative amoebiasis therapies.

Several efforts to find such alternative therapies have been undertaken over the years, including the recent development of the antirheumatic drug auranofin as a promising potential treatment for amoebiasis [14–16]. One noteworthy direction of anti-amoebic drug research has been the efforts of multiple groups to inhibit *E. histolytica* by targeting specific kinase proteins believed to be critical to the parasite's functioning [17–19]. This approach has often involved computational modeling and *in-silico* screening of compounds against the kinases of interest, followed by *in-vitro* tests of top-scoring molecules [17]. These efforts have resulted in both the discovery of potent new hit compounds as well as validation of the previously discovered activity of auranofin [17]. However, despite these successes, no new clinical treatments have yet been produced.

Importantly, one promising area currently unexplored by such studies is the potential of existing human kinase inhibitor drugs. A particular advantage of these drugs is the rich array

of data available regarding their activity profiles against human target proteins, which allows for the mapping and utilization of their complex multi-target pharmacology. Such maps could in turn be projected into the *E. histolytica* proteome and used to infer potential antiamoebic drug activity by identifying drugs with similar target profiles to known active compounds. We have previously published a computational tool capable of such mapping for antineoplastic drugs, including a large number of kinase inhibitors [20]. We describe here the use of this tool to prioritize molecules for screening against *E. histolytica* trophozoites based on initial hits from a small primary screen. In total, 6 antineoplastic kinase inhibitors (AKIs) were found to have potent and rapid anti-amoebic activity. The results of these experiments demonstrate the promise of using target-based analysis to leverage compounds with multi-target pharmacology against a human parasite.

Materials and methods

E. histolytica cell culture

E. histolytica strain HM-1:IMSS trophozoites were maintained in 50ml culture flasks (Greiner Bio-One) containing TYI-S-33 media, 10% heat-inactivated adult bovine serum (Sigma), 1% MEM Vitamin Solution (Gibco), supplemented with penicillin (100 U/mL) and streptomycin (100 μ g/mL) (Omega Scientific) [14].

Compounds

Compounds for screens were purchased from Fisher Scientific and Millipore-Sigma.

Cell viability screen to determine drug potency against E. histolytica

Following a previously-published approach [14] E. histolytica trophozoites maintained in the logarithmic phase of growth were seeded into 96-well plates (Greiner Bio-One) at 5,000 cells/ well to a total volume of 100 µl/well. 8- or 16-point two-fold dilution series of the treatment compounds were prepared, beginning at a maximum final treatment concentration of 50 µM. $0.5 \,\mu$ l of each drug concentration was added to triplicate wells for each treatment group. $0.5 \,\mu$ l of DMSO was used as a negative control, and 0.5 μ l of 10 mM metronidazole dissolved in DMSO was used as a positive control, giving a final concentration of 50 μ M. Alternatively, wells with only media were used as a negative control. The plates were placed in GasPak EZ (Becton-Dickinson) bags and incubated at 37°C for 48hr. Plates were removed and 50 µl of CellTiter-Glo (Promega) was added to each well. Plates were shaken and incubated in darkness for 20 minutes and the luminescence value of each well was read by a luminometer (EnVision, PerkinElmer). Percent inhibition was calculated by subtracting the luminescence values of each experimental data point from the average minimum signal obtained from positive control values and dividing by the difference between the average maximum signal negative control and the positive control. The resulting decimal value was then multiplied by 100 to give a percentage.

Determination of drug EC₅₀ values in vitro over time

Effects of different concentrations of compounds on *E. histolytica* trophozoite cell viability were determined as described in the previous section at a series of timepoints ranging from 6 hours to 48 hours following drug administration. EC_{50} values were calculated at each timepoint as previously described.

Determination of varying drug exposure time effects

E. histolytica trophozoites were treated with either 5 μ M ponatinib, 5 μ M neratinib, 5 μ M olmutinib, or 10 μ M metronidazole in replicates of 4 wells in 96-well plates. At timepoints ranging from 2 to 48 hours, wells were aspirated, washed once with fresh media, and refilled with fresh media. At 48 hours, percentage trophozoite inhibition was measured using luminescence and calculated as described previously for each timepoint.

Identification of desired target profile of active drugs

A desired target profile of active drugs was generated using the "Multi-drug target finder" tool in the CancerDrugMap (http://ruben.ucsd.edu/dnet/maps/drug_find.html) [20]. Drugs that were active (dasatinib, bosutinib, ibrutinib. . .) and inactive (nilotinib, imatinib. . .) in the *E. histolytica* proliferation assay were inputs, respectively. Drug-target interaction activity data for the tool were collected from multiple sources as previously described, including ChEMBL, PubChem, and literature sources [20]. Drug targets were ranked based on the drug-target activity data and using the following equations and assembled into the final anti-amoebic activity-associated profile.

Where: Score of target $S = \sum_{drugs} weight \times (pAct-4)$

$$pAct = -log(IC50/Kd/Ki)$$

For active drugs: *weight* = 1

For inactive drugs: weight $= -0.6 \times \sqrt{\frac{number of active drugs}{number of inactive drugs}}$

Identification of drugs with desired target profile

A list of drugs with target profiles matching the desired target profile was generated with the "Multi-target drug finder" tool in CancerDrugMap (http://ruben.ucsd.edu/dnet/maps/tar_find.html). The desired target profiles generated above were input correspondingly. Resulting cancer drugs were ranked based on the drug-target activity and the following equations.

Score of drug
$$S = \sum_{target} weight \times (pAct - 4)$$

Where

$$pAct = -log(IC50/Kd/Ki)$$

For targets to hit: weight = 1

For targets to avoid: weight = $-0.6 \times \sqrt{\frac{number of wanted targets}{number of unwanted targets}}$

Top-ranking drugs were then selected for further testing.

Identification of potential targets of active drugs in the *E. histolytica* genome

15 top ranking human protein targets (YES1, ABL1, BTK, BMX, LCK, HCK, FGR, BLK, ERBB4, LYN, FYN, SRC, CSK, ABL2, and FRK) from the identified desired target profile were searched against the *E. histolytica* genome downloaded from (https://amoebadb.org/common/ downloads/Current_Release/EhistolyticaHM11MSS/). Human protein sequences were downloaded from (uniprot.org) and the annotated kinase domains of each protein were compiled into a single file. Full gapped optimal sequence alignments with zero end gap penalties (ZEGA) were performed between the kinase domain sequences of the 15 targets and the *E. his-tolytica* genome. The significance of each alignment was assessed according to a number of residue substitution matrices as a pP value (pP = -log(P-value)) [21].

E. histolytica genes with pP over 10, namely the P-value of the alignment lower than 10^{-10} were selected as potential targets. A network map of the 15 top ranking human genes and homologous *E. histolytica* genes was generated with Graphviz neato, with edges corresponding to the pP values between human and *E. histolytica* genes.

Cyst killing assay

For assays on mature cysts, a transgenic *E. invadens* line stably expressing luciferase (CK-luc) was used [22]. Mature cyst viability assay was performed as described previously [16]. Parasites were induced to encyst by incubation in encystation media (47% LG) [23]. After 72 h, parasites were washed once in distilled water and incubated at 25°C for 4–5 h in water to lyse trophozoites. Purified cysts were pelleted, counted to ensure equal cyst numbers, and resuspended in encystation media at a concentration of $1-5x10^5$ cells per ml. One ml suspension per replicate was transferred to glass tubes containing encystation media and drug or DMSO, then incubated at 25°C for 72 h. On the day of the assay, cysts were pelleted and treated once more with distilled water for 5 h to lyse any trophozoites that had emerged during treatment. Purified cysts were then resuspended in 75 µl Cell Lysis buffer (Promega) and sonicated for 2x10 seconds to break the cyst wall. Luciferase assay was performed using the Promega luciferase assay kit according to the manufacturer's instructions. Assays were performed on equal volume of lysate (35 µl) and not normalized to protein content. Effect of the drug was calculated by comparison to DMSO control, after subtraction of background signal. Significance of drug effects was calculated using a one-tailed T-test.

Results

Screen of antineoplastic kinase inhibitors against E. histolytica trophozoites

In order to identify anti-amoebic activity among antineoplastic kinase inhibitors, a selection of 12 drugs was screened against *E. histolytica* trophozoites *in vitro*. All compounds tested were FDA-approved cancer chemotherapy drugs designed to inhibit human kinase proteins as their mechanism of action. *E. histolytica* trophozoites were seeded into 96-well plates along with a serially-diluted range of drug concentrations. Trophozoites were incubated for 48 hours, after which the surviving cell amount was determined using a luciferase-based cell viability assay. Percent inhibition of trophozoite growth was calculated for each treatment well in comparison with vehicle-only negative controls representing 0% inhibition. From this data EC₅₀ values were calculated for each respective drug (Table 1). Out of the 12 drugs tested, ibrutinib, dasatinib, and bosutinib all were found to possess EC₅₀ values similar to or lower than the EC₅₀ values of of 2–5 μ M for the currently used drug metronidazole (Fig 1). Based on these results we concluded that antineoplastic kinase inhibitor drugs are capable of potent inhibition of *E. histolytica* and determined that further analysis and refinement was warranted in order to discover even more potent drugs in the same class.

Analysis of hits

In order to identify potent anti-amoebic candidates from the existing pool of AKI drugs, an approach was utilized where human protein target profiles were computationally generated for drugs active in the screen, followed by identification of additional drugs with matching or

Table 1. Results of primary screen of AKI drugs against *E. histolytica* **trophozoites**. Color scale indicates drug potency: darker blue = more potent, lighter blue = less potent. Anti-amoebic activity classified based on EC₅₀ value as: Very high (0.001–0.999 μ M), High (1.000–4.999 μ M), Moderate (5.000–9.999 μ M), Low (10.000–19.999 μ M), Very low (20.000–99.999 μ M), or None (EC₅₀ > 100.000 μ M).

Drug name	EC ₅₀ (μM)	Anti-amoebic activity
Ibrutinib	0.98	Very high
Dasatinib	1.57	High
Bosutinib	1.94	High
Nilotinib	8.22	Moderate
Gefitinib	8.52	Moderate
Sunitinib	10.09	Low
Afatinib	11.69	Low
Crizotinib	12.83	Low
Erlotinib	42.07	Very low
Dabrafenib	55.18	Very low
Vemurafenib	>100	None
Imatinib	>100	None

https://doi.org/10.1371/journal.pntd.0008425.t001

similar target profiles. To generate the target profiles for active drugs, a computational tool called CancerDrugMap (CDM) was utilized, which we have previously described, and which is available at: *ruben.ucsd.edu/dnet/* [20]. Using CDM, human targets of both the active and inactive compounds from the initial were compared. Protein targets were scored and ranked based on targeting activity of active compounds as well as lack of targeting activity by the inactive ones. This generated a profile of human protein targets associated with the active amoebicidal drugs in the screen (Fig 2). Based on this profile, two strategies were then employed to identify drugs with similar target profiles and hence the potential for similar anti-amoebic activity (Fig 3). In the first strategy, CDM was used to score and rank all AKI drugs in the database based on their activity against the complete ranked target profile, using an algorithm described in Materials and Methods. Drugs possessing a score greater than or equal to 15 were selected for



Fig 1. EC₅₀ determination of antineoplastic kinase inhibitors against *E. histolytica* trophozoites. Dose response curve plotting percentage inhibition of *E. histolytica* trophozoites compared to drug concentrations of antineoplastic kinase inhibitors. Trophozoites were assayed for cell viability following treatment with each drug for a period of 48 hours. The three drugs with the lowest EC₅₀ values (ibrutinib, dasatinib, and bosutinib) are plotted in red, purple, and blue. All drugs with EC₅₀ values > 2 μ M are plotted in gray. Each data point represents mean values of percentage inhibition. Error bars represent standard deviation. Complete list of drugs can be found in Table 1.

https://doi.org/10.1371/journal.pntd.0008425.g001





further *in vitro* testing (Fig 4). This cutoff value was chosen based on a naturally occurring gap between drugs with a score of 15 and the next-lowest scoring drugs, with the latter possessing only few and very weak activity scores against any of the target profile proteins. The second strategy was identical to the first with the exception that CDM was used to score and rank drugs based on activity against target proteins individually rather than the complete profile. Drugs active above a threshold score of 15 against individual proteins from the target profile were ranked by the number of proteins from the profile they possessed this level of activity against (Table 2). Drugs possessing a score at or over the threshold of 15 against more than 2 proteins from the target profile were selected for further investigation. The first and second strategies combined generated a list of 15 drugs with similar known human protein targets to the positive hits from the initial screen, and hence high potential for corresponding anti-amoebic activity.



Fig 3. Graphical screening workflow of antineoplastic kinase inhibitors against *E. histolytica*. Chemical structures represent the three drugs found to possess the lowest EC_{50} values in each screen (ibrutinib, dasatinib, bosutinib, and ponatinib, neratinib, olmutinib respectively).

https://doi.org/10.1371/journal.pntd.0008425.g003

Potential drug target proteins are present in the E. histolytica proteome

While the data regarding activity of AKI drugs against human target proteins is valuable for the purpose of grouping drugs with the potential for similar activity against *E. histolytica*, it does not provide information regarding the drugs' actual protein targets in the parasite. In





order to identify whether potential drug targets exist in *E. histolytica* that are similar to the human target profile proteins, the human proteins were searched against the *E. histolytica* proteome. In order to do so, the kinase domain sequences of the human proteins were extracted and aligned against the complete published set of *E. histolytica* open reading frames (ORFs). 32 *E. histolytica* ORFs were found to align to the human sequences with a p-value of 10^{-10} or less. A network map was generated of these top-scoring *E. histolytica* proteins and their relationship to the human protein targets (Fig 5). In the network map multiple *E. histolytica* ORFs can be seen to possess strong alignments to several human sequences. These results demonstrate the possibility that *E. histolytica* may possess protein targets equivalent to those known to be targeted in humans by the active AKI drugs.

Extended screen of candidate drugs based on primary analysis

Based on our CDM analysis we tested the list of 12 potentially active AKI drugs against *E. his-tolytica* trophozoites in an extended *in vitro* screen. Compounds were tested as previously, using the same luciferase-based cell viability assay to determine EC_{50} values. The drugs ponatinib, neratinib, and olmutinib were found to possess highly potent activity in this screen, all with sub-micromolar EC_{50} values (Table 3) (Fig 6).

Hit compounds kill E. histolytica trophozoites

An important question regarding the activity of any compound intended to act against *E. his-tolytica* is whether it induces cell death in the parasite or merely slows its replication. In order to determine which type of activity belongs to each of the AKI drugs active in the initial and extended screens, we measured the number of surviving cells after 48 hours of drug treatment compared to freshly-counted aliquots of cells. 5,000 cells per well were seeded into 96-well

Table 2. Analysis of drugs based on activity towards individual proteins in the active drug target profile. Drugs are ranked based on the number of proteins from the active drug target profile towards which they possess an activity score greater than a threshold value. Darker color indicates a greater number of proteins. Drugs possessing the desired level of activity towards more than two target profile proteins were considered for further screening, shown above red line.

Drug name	Number of target profile matches
Dasatinib	15
Bosutinib	15
Ponatinib	13
Ibrutinib	11
Vandetanib	6
Cediranib	6
Sunitinib	4
Nilotinib	4
Masitinib	4
Regorafenib	3
Nintedanib	3
Neratinib	3
Imatinib	3
Brigatinib	3
Acalabrutinib	3
Sorafenib	2
Olmutinib	2
Erlotinib	2
Crizotinib	2
Axitinib	2
Rociletinib	1
Osimertinib	1
Lapatinib	1
Gefitinib	1
Afatinib	1

https://doi.org/10.1371/journal.pntd.0008425.t002

plates and treated with dasatinib, bosutinib, ibrutinib, ponatinib, neratinib, olmutinib, metronidazole, or vehicle. A concentration of 10μ M was used for all drugs in order to ensure maximal and unambiguous inhibitory effects against the trophozoites. After 48 hours fresh aliquots containing a known number of cells were seeded into empty wells, CellTiter-Glo was added, and the luminescence of all wells was measured. Using the linear relationship of CellTiter-Glo luminescence to the number of cells being assayed, the number of cells in treatment group wells was calculated using their luminescence values relative to those of the freshly-aliquoted wells. All drugs tested were found to have significantly decreased the number of live cells in their treatment groups below the initial 5,000 cells. In contrast, cells treated with only vehicle significantly increased in number to over 14,000 cells per well (Fig 7). Interestingly, none of the drugs tested, including metronidazole, completely reduce the estimated cell number to zero. This could possibly be due to either a small number of genuine surviving cells, or to dying cells which still contain measurable ATP.

In order to characterize whether the active drugs genuinely kill *E. histolytica* trophozoites or merely act as false positives by inhibiting the ATP-driven, luciferase-based CellTiter-Glo assay system we tested concentration ranges of each drug on cells and immediately after

PLOS NEGLECTED TROPICAL DISEASES



Fig 5. Network map of active drug profile proteins with orthologous *E. histolytica* ORFs. Orange ovals represent human sequences. Blue ovals represent *E. histolytica* sequences. Lines represent alignment relationships possessing a calculated pP greater than the cutoff value of 10. Line color represents pP value, with darker lines denoting higher pP. Blue ovals shown in the center with the highest number of connecting lines represent the most likely *E. histolytica* orthologs of the human target proteins.

https://doi.org/10.1371/journal.pntd.0008425.g005

addition of drugs. If the drugs were acting on the CellTiter-Glo assay reagents rather than the cells themselves, a dose-response relationship of drug to assay activity should have been evident. However, no dose-response relationship was observed, and measured luminescence values remained equivalent across all concentrations of drug treatments (S1 Fig). These results indicate that the active drugs are true positives against *E. histolytica* cells and do not inhibit the assay itself.

Table 3. Results of extended screen of AKI drugs against *E. histolytica* trophozoites. Color scale indicates drug potency: darker blue = more potent, lighter blue = less potent. Anti amoebic activity classified based on EC₅₀ value as: Very high (0.001–0.999 μM), High (1.000–4.999 μM), Moderate (5.000–9.999 μM), Low (10.000–19.999 μM), Very low (20.000–99.999 μM), or None (EC₅₀ > 100.000 μM).

Drug name	EC ₅₀ (μM)	Anti-amoebic activity
Ponatinib	0.1299	Very high
Neratinib	0.3113	Very high
Olmutinib	0.6462	Very high
Nintedanib	4.239	High
Cediranib	7.286	Moderate
Vandetanib	8.971	Moderate
Acalabrutinib	11.34	Low
Masitinib	14.03	Low
Regorafenib	15.94	Low
Sorafenib	18.3	Low
Pazopanib	>100	None
Axitinib	>100	None

https://doi.org/10.1371/journal.pntd.0008425.t003

Hit compounds kill E. histolytica trophozoites as rapidly as metronidazole

In addition to drug potency, an important characteristic of any drug is the rapidity with which it achieves its desired effect. In order to characterize this aspect of the most active drugs from the trophozoite viability screens, the EC_{50} values of ponatinib, neratinib, olmutinib, and metronidazole against *E. histolytica* trophozoites were measured at a series of timepoints after treatment initiation. The luciferase-based CellTiter-Glo cell viability assay was used as previously to determine the percent inhibition in each set of experimental replicates. Duplicate plates containing cells treated with serially-diluted ranges of ponatinib, neratinib, olmutinib, and metronidazole concentrations were prepared for each desired time point. Measurements were collected at 12, 24, 36, and 48 hours post-drug-treatment respectively. From the data



Fig 6. Extended screen of antineoplastic kinase inhibitors against *E. histolytica* trophozoites. Dose response curves plotting percentage inhibition of *E. histolytica* trophozoites at different drug concentrations. Trophozoites were assayed for cell viability following treatment with each drug for a period of 48 hours. The three drugs with the lowest EC_{50} values (ponatinib, neratinib, and olmutinib) are plotted in red, purple, and blue. All drugs with EC_{50} values > 2 µM are plotted in gray. Each data point represents mean values. Error bars represent standard deviation. Complete list of drugs tested found in Table 3.

https://doi.org/10.1371/journal.pntd.0008425.g006



Fig 7. Amoebicidal effects of antineoplastic kinase inhibitor drugs. Live cell number calculated in comparison to aliquots of known amounts of cells. All drugs tested at 10μ M, vehicle = 0.5% DMSO. Dotted line represents the 5000 cells originally seeded into all wells for each treatment group. Error bars represent standard deviation.

https://doi.org/10.1371/journal.pntd.0008425.g007

obtained EC_{50} values were calculated for each time point of each drug treatment and compared over time (Fig 8A, 8B, 8C and 8D). All drugs tested achieved steady EC_{50} values within 36 hours equivalent to those observed at 48 hours (Fig 8E). These results indicate that the active AKI drugs achieve their anti-amoebic effects as rapidly as the current treatment, metronidazole.

Hit compounds and metronidazole require similar exposure times for *E*. *histolytica* trophozoite inhibition

To determine the amount of exposure time necessary for parasite killing by the compounds most active in the initial and extended screens, E. histolytica trophozoite inhibition was measured following varying treatment periods with either ponatinib, neratinib, olmutinib, or metronidazole. Cells were treated with drugs at 5 µM for intervals ranging from 2 to 48 hours, followed by drug washout and continued incubation to 48 total hours. This concentration was chosen to ensure complete parasite killing by all drugs after 48 hours. Trophozoite viability was measured for each drug and exposure time using CellTiter-glo and the percentage inhibition was calculated. Because each drug tested possess a different anti-amoebic EC₅₀ value, the percentage inhibition for each exposure time was scaled to the 48-hour value for each respective drug. As a result, the relative effectiveness of varying exposure times could be compared across drugs irrespective of varying drug potency. All drug treatments achieved inhibition levels after 24 hours of exposure time similar to levels observed after 48 hours (Fig 8F). These results indicate that ponatinib, neratinib, olmutinib, and metronidazole all require roughly 24 hours of parasite exposure time in order to achieve maximal levels of parasite inhibition. Additionally, both ponatinib and neratinib achieved significantly higher scaled % inhibition levels after 2, 6, and 12 hours compared to metronidazole, indicating that a somewhat shorter exposure time might be required for these drugs to achieve their antiparasitic effects.

Antineoplastic kinase inhibitors kill mature Entamoeba cysts

A major drawback of metronidazole as a treatment for amebiasis is its poor activity against luminal parasites and cysts [2]. To determine if AKIs may be superior in this respect, we



Fig 8. Timing of drug action against *E. histolytica* trophozoites. (A—D) Dose-response curves measured at 12, 24, 36, and 48hr for ponatinib, neratinib, olmutinib, and metronidazole. (E) Plot of EC_{50} values calculated from the data shown in (A-D) graphed over time. (F) Plot of *E. histolytica* trophozoite inhibition by AKI drugs after varying exposure times. Trophozoites were treated with ponatinib, neratinib, olmutinib, or metronidazole at 5 μ M for 2, 6, 12, 24, 36, or 48hr, followed by drug washout and continued incubation until 48hr. Percent inhibition was then determined. Points represent % inhibition for each drug and exposure time scaled to the 48hr % inhibition value for the same drug. Statistical difference between groups was determined by 1-way ANOVA for each exposure time. (** = p < 0.01) (* = p < 0.05).

assayed for killing of mature *Entamoeba* cysts. As *E. histolytica* cannot be induced to encyst *in vitro* [23], the related parasite, *E. invadens*, a well-characterized model system for *Entamoeba* development, was utilized. Mature (72h) cysts of a transgenic line constitutively expressing luciferase were treated with 10 μ M dasatinib, bosutinib, ibrutinib, or 0.5% DMSO as negative control, for 3 days. After treatment, cysts were treated with distilled water for five hours to remove any remaining trophozoites, and luciferase activity was assayed. Ibrutinib was found to significantly reduce luciferase signal to between 10% and 50% of controls, indicating that this AKI drug is capable of killing *Entamoeba* cysts. In contrast, metronidazole up to 20 μ M had no effect (Fig 9). As ibrutinib is known to act as a covalent inhibitor of human kinase proteins, another such covalent inhibitor which showed activity in the extended screen, acalabrutinib, was also tested [24,25]. However, this drug did not consistently display any significant cysticidal activity.





Discussion

Treatment options for amoebiasis are currently limited to either nitroimidazole drugs such as metronidazole, which acts via anaerobic activation to toxic reactive forms in *E. histolytica* [10]. While other drugs have been proposed or used at times, a 2013 systematic review concluded that only nitroimidazole drugs and the thiazolide drug nitazoxanide are likely to be beneficial to patients [10]. This fact, when coupled with emerging drug resistance to metronidazole as well as its lack of activity against the infectious cyst form of *E. histolytica* necessitates the search for new treatment options [13].

In this study we tested the hypothesis that E. histolytica could be killed by FDA-approved antineoplastic kinase inhibitors, possibly via action on parasitic homologs of human kinases. Out of 24 such drugs tested, six were shown to possess strong anti-amoebic properties, representing a completely new class of drugs in this area. All of the six highly active drugs displayed unique and important advantages over the current treatment. Dasatinib, ibrutinib, bosutinib, ponatinib, neratinib, and olmutinib were all for the first time shown to kill E. histolytica trophozoites in vitro significantly more potently than metronidazole. Ponatinib, neratinib, and olmutinib in particular demonstrated sub-micromolar EC₅₀ values rarely observed for any compound against this organism. These latter three were also shown to act as rapidly as metronidazole, and two of them, ponatinib and neratinib, were shown to act after shorter exposure times. Significantly, ibrutinib was also shown to kill the cysts of the related model organism E. invadens in contrast to metronidazole which was not. This feature is particularly unique and desirable for epidemiological purposes. Outside of the current study, both ibrutinib and neratinib have been shown to possess good brain penetrance, as does metronidazole [26-28]. Taken together all these properties give the six drugs strong potential for repurposing against E. histo*lytica* infection, especially in advanced amoebiasis cases where infection has progressed to the liver or brain, or in cases of cancer patients with *E. histolytica* infections [3,8,29].

Interestingly, a recent high-throughput screen of the reframeDB commercial drug library also observed activity of ponatinib and dasatinib against *E. histolytica*, as well as the tyrosine kinase inhibitor rebastinib not included in the current study. While this screen is currently unpublished, the results can be viewed at (https://reframedb.org/assays/A00203). The

researchers involved in this screen have also conducted screens of the same library against the parasitic amoebae *Naegleria fowleri* and *Balamuthia mandrillaris*, both of which found ponatinib to be active [30]. All of these results further validate the potential for AKIs as a new class of highly potent drugs against *E. histolytica*, and potentially other parasitic amoeba as well.

One drawback to AKIs which might limit their use as hypothetical clinical antiparasitic drugs is their ability to cause moderate adverse effects in humans. All of the drugs found to be highly active in this study are known to possess this downside [31–40]. This has been observed both in studies of cancer patients and of healthy volunteers, in which these drugs are nevertheless generally described as "safe and well-tolerated" [41–44]. In particular, AKIs tend to cause diarrhea as one of the most common adverse effects and as such have the potential to exacerbate the symptoms of a diarrheal disease such as amoebiasis [37–39]. This could possibly be due to disruption of the intestinal microbiome by the AKI drugs, which has been previously reported [45,46]. It is not noting however, that this feature is shared in common with the current standard of care for amoebiasis, metronidazole [7]. As such it may not necessarily disqualify AKIs from use against this disease, especially as it has been found to be easily manageable in clinical trials with standard anti-diarrheal therapy [38]. Another strategy to circumvent the adverse effects associated with these AKIs could involve the testing of structurally related molecules for activity against amoebae without activity against human kinases.

Taken together the results of this study document a new class of FDA-approved drugs with strong potential for repurposing against a widespread and devastating pathogen. Future research may expand on these findings by characterizing the molecular mechanisms underlying the actions of these drugs as well as testing their *in vivo* efficacy.

Supporting information

S1 Fig. False-positive assay against *E. histolytica* **trophozoites.** All drugs were tested at a serially-diluted range of concentrations. Cell viability measured at T = 0. (TIF)

S1 Dataset. Fig 1 Data. (XLSX)

S2 Dataset. Fig 6 Data. (XLSX)

S3 Dataset. Fig 7 Data. (XLSX)

S4 Dataset. Fig 8 Data. (XLSX)

S5 Dataset. Fig 9 Data. (XLSX)

S6 Dataset. S1 Fig Data. (XLSX)

Acknowledgments

The authors of this study would like to thank Dr. Jim McKerrow and the Center for Discovery and Innovation in Parasitic Diseases at the Skaggs School of Pharmacy at the University of California—San Diego as well as Lily Hahn, Thi Nguyen, and Abdolhakim Mohammed for their contributions to this work.

Author Contributions

Conceptualization: Conall Sauvey, Ruben Abagyan.

Data curation: Conall Sauvey.

Formal analysis: Conall Sauvey, Da Shi.

Funding acquisition: Ruben Abagyan.

Investigation: Conall Sauvey, Gretchen Ehrenkaufer, Da Shi.

Methodology: Conall Sauvey, Gretchen Ehrenkaufer, Da Shi, Anjan Debnath.

Project administration: Conall Sauvey.

Resources: Ruben Abagyan.

Software: Da Shi.

Supervision: Anjan Debnath, Ruben Abagyan.

Visualization: Conall Sauvey.

Writing - original draft: Conall Sauvey.

Writing – review & editing: Gretchen Ehrenkaufer, Da Shi, Anjan Debnath, Ruben Abagyan.

References

- Shirley DT, Farr L, Watanabe K, Moonah S. A Review of the Global Burden, New Diagnostics, and Current Therapeutics for Amebiasis. Open forum infectious diseases. 2018; 5(7):ofy161. <u>https://doi.org/10.1093/ofid/ofy161</u> PMID: 30046644
- Pritt BS, Clark CG. Amebiasis. Mayo Clin Proc. 2008; 83(10):1154–9. quiz 9–60. https://doi.org/10. 4065/83.10.1154 PMID: 18828976
- Ralston KS, Petri WA Jr. Tissue destruction and invasion by Entamoeba histolytica. Trends Parasitol. 2011; 27(6):254–63. https://doi.org/10.1016/j.pt.2011.02.006 PMID: 21440507
- Faria CP, Zanini GM, Dias GS, da Silva S, de Freitas MB, Almendra R, et al. Geospatial distribution of intestinal parasitic infections in Rio de Janeiro(Brazil) and its association with social determinants. PLoS Negl Trop Dis. 2017; 11(3):e0005445. https://doi.org/10.1371/journal.pntd.0005445 PMID: 28273080
- Sahimin N, Lim YA, Ariffin F, Behnke JM, Lewis JW, Mohd Zain SN. Migrant Workers in Malaysia: Current Implications of Sociodemographic and Environmental Characteristics in the Transmission of Intestinal Parasitic Infections. PLoS Negl Trop Dis. 2016; 10(11):e0005110. https://doi.org/10.1371/journal. pntd.0005110 PMID: 27806046
- Verkerke HP, Petri WA Jr, Marie CS. The dynamic interdependence of amebiasis, innate immunity, and undernutrition. Semin Immunopathol. 2012; 34(6):771–85. https://doi.org/10.1007/s00281-012-0349-1 PMID: 23114864
- Dusengeyezu E, Kadima J. How do Metronidazole Drawbacks Impact Patient Compliance and Therapeutic Outcomes in Treating Amoebiasis in Rwanda. International Journal of TROPICAL DISEASE & Health. 2016; 17(3):1–7.
- Carrero JC, Reyes-Lopez M, Serrano-Luna J, Shibayama M, Unzueta J, Leon-Sicairos N, et al. Intestinal amoebiasis: 160 years of its first detection and still remains as a health problem in developing countries. Int J Med Microbiol. 2020; 310(1):151358. https://doi.org/10.1016/j.ijmm.2019.151358 PMID: 31587966
- Garduno-Espinosa J, Martinez-Garcia MC, Fajardo-Gutierrez A, Ortega-Alvarez M, Alvarez-Espinosa A, Vega-Perez V, et al. Frequency and risk factors associated with metronidazole therapeutic noncompliance. Revista de investigacion clinica; organo del Hospital de Enfermedades de la Nutricion. 1992; 44(2):235–40. PMID: 1439312
- 10. Marie C, Petri WA Jr. Amoebic dysentery. BMJ clinical evidence. 2013;2013. PMID: 23991750
- Kikuchi T, Koga M, Shimizu S, Miura T, Maruyama H, Kimura M. Efficacy and safety of paromomycin for treating amebiasis in Japan. Parasitol Int. 2013; 62(6):497–501. <u>https://doi.org/10.1016/j.parint.</u> 2013.07.004 PMID: 23850836

- Blessmann J, Tannich E. Treatment of asymptomatic intestinal Entamoeba histolytica infection. N Engl J Med. 2002; 347(17):1384. https://doi.org/10.1056/NEJM200210243471722 PMID: 12397207
- Wassmann C, Hellberg A, Tannich E, Bruchhaus I. Metronidazole resistance in the protozoan parasite Entamoeba histolytica is associated with increased expression of iron-containing superoxide dismutase and peroxiredoxin and decreased expression of ferredoxin 1 and flavin reductase. J Biol Chem. 1999; 274(37):26051–6. https://doi.org/10.1074/jbc.274.37.26051 PMID: 10473552
- Debnath A, Parsonage D, Andrade RM, He C, Cobo ER, Hirata K, et al. A high-throughput drug screen for Entamoeba histolytica identifies a new lead and target. Nat Med. 2012; 18(6):956–60. https://doi. org/10.1038/nm.2758 PMID: 22610278
- Bashyal B, Li L, Bains T, Debnath A, LaBarbera DV. Larrea tridentata: A novel source for anti-parasitic agents active against Entamoeba histolytica. Giardia lamblia and Naegleria fowleri PLoS Negl Trop Dis. 2017; 11(8):e0005832. https://doi.org/10.1371/journal.pntd.0005832 PMID: 28793307
- Ehrenkaufer GM, Suresh S, Solow-Cordero D, Singh U. High-Throughput Screening of Entamoeba Identifies Compounds Which Target Both Life Cycle Stages and Which Are Effective Against Metronidazole Resistant Parasites. Front Cell Infect Microbiol. 2018; 8:276. https://doi.org/10.3389/fcimb.2018. 00276 PMID: 30175074
- Mi-Ichi F, Ishikawa T, Tam VK, Deloer S, Hamano S, Hamada T, et al. Characterization of Entamoeba histolytica adenosine 5'-phosphosulfate(APS) kinase; validation as a target and provision of leads for the development of new drugs against amoebiasis. PLoS Negl Trop Dis. 2019; 13(8):e0007633. https:// doi.org/10.1371/journal.pntd.0007633 PMID: 31425516
- Nurkanto A, Jeelani G, Yamamoto T, Naito Y, Hishiki T, Mori M, et al. Characterization and validation of Entamoeba histolytica pantothenate kinase as a novel anti-amebic drug target. Int J Parasitol Drugs Drug Resist. 2018; 8(1):125–36. https://doi.org/10.1016/j.ijpddr.2018.02.004 PMID: 29518650
- Anamika K, Bhattacharya A, Srinivasan N. Analysis of the protein kinome of Entamoeba histolytica. Proteins. 2008; 71(2):995–1006. https://doi.org/10.1002/prot.21790 PMID: 18004777
- Shi D, Khan F, Abagyan R. Extended Multitarget Pharmacology of Anticancer Drugs. J Chem Inf Model. 2019; 59(6):3006–17. https://doi.org/10.1021/acs.jcim.9b00031 PMID: 31025863
- Abagyan RA, Batalov S. Do aligned sequences share the same fold? J Mol Biol. 1997; 273(1):355–68. https://doi.org/10.1006/jmbi.1997.1287 PMID: 9367768
- Ehrenkaufer GM, Singh U. Transient and stable transfection in the protozoan parasite Entamoeba invadens. Mol Biochem Parasitol. 2012; 184(1):59–62. <u>https://doi.org/10.1016/j.molbiopara.2012.04.007</u> PMID: 22561071
- Sanchez L, Enea V, Eichinger D. Identification of a developmentally regulated transcript expressed during encystation of Entamoeba invadens. Mol Biochem Parasitol. 1994; 67(1):125–35. <u>https://doi.org/10.1016/0166-6851(94)90102-3 PMID: 7838173</u>
- Barf T, Covey T, Izumi R, van de Kar B, Gulrajani M, van Lith B, et al. Acalabrutinib(ACP-196): A Covalent Bruton Tyrosine Kinase Inhibitor with a Differentiated Selectivity and In Vivo Potency Profile. J Pharmacol Exp Ther. 2017; 363(2):240–52. https://doi.org/10.1124/jpet.117.242909 PMID: 28882879
- 25. Davids MS, Brown JR. Ibrutinib: a first in class covalent inhibitor of Bruton's tyrosine kinase. Future Oncol. 2014; 10(6):957–67. https://doi.org/10.2217/fon.14.51 PMID: 24941982
- 26. Bradley CA. Brain metastases respond to neratinib plus capecitabine. Nat Rev Clin Oncol. 2019; 16 (6):336.
- Goldwirt L, Beccaria K, Ple A, Sauvageon H, Mourah S. Ibrutinib brain distribution: a preclinical study. Cancer Chemother Pharmacol. 2018; 81(4):783–9. https://doi.org/10.1007/s00280-018-3546-3 PMID: 29476222
- Duchnowska R, Loibl S, Jassem J. Tyrosine kinase inhibitors for brain metastases in HER2-positive breast cancer. Cancer Treat Rev. 2018; 67:71–7. https://doi.org/10.1016/j.ctrv.2018.05.004 PMID: 29772459
- 29. Petri WA, Haque R. Entamoeba histolytica brain abscess. Handb Clin Neurol. 2013; 114:147–52. https://doi.org/10.1016/B978-0-444-53490-3.00009-1 PMID: 23829905
- 30. Kangussu-Marcolino MM, Ehrenkaufer GM, Chen E, Debnath A, Singh U. Identification of plicamycin, TG02, panobinostat, lestaurtinib, and GDC-0084 as promising compounds for the treatment of central nervous system infections caused by the free-living amebae Naegleria, Acanthamoeba and Balamuthia. Int J Parasitol Drugs Drug Resist. 2019; 11:80–94. https://doi.org/10.1016/j.ijpddr.2019.10.003 PMID: 31707263
- Kantarjian HM, Cortes JE, Kim DW, Khoury HJ, Brummendorf TH, Porkka K, et al. Bosutinib safety and management of toxicity in leukemia patients with resistance or intolerance to imatinib and other tyrosine kinase inhibitors. Blood. 2014; 123(9):1309–18. <u>https://doi.org/10.1182/blood-2013-07-513937</u> PMID: 24345751

- Kaur V, Swami A. Ibrutinib in CLL: a focus on adverse events, resistance, and novel approaches beyond ibrutinib. Ann Hematol. 2017; 96(7):1175–84. <u>https://doi.org/10.1007/s00277-017-2973-2</u> PMID: 28342031
- Dorer DJ, Knickerbocker RK, Baccarani M, Cortes JE, Hochhaus A, Talpaz M, et al. Impact of dose intensity of ponatinib on selected adverse events: Multivariate analyses from a pooled population of clinical trial patients. Leuk Res. 2016; 48:84–91. <u>https://doi.org/10.1016/j.leukres.2016.07.007</u> PMID: 27505637
- 34. Ottmann O, Saglio G, Apperley JF, Arthur C, Bullorsky E, Charbonnier A, et al. Long-term efficacy and safety of dasatinib in patients with chronic myeloid leukemia in accelerated phase who are resistant to or intolerant of imatinib. Blood Cancer J. 2018; 8(9):88. https://doi.org/10.1038/s41408-018-0122-3 PMID: 30190469
- Caldemeyer L, Dugan M, Edwards J, Akard L. Long-Term Side Effects of Tyrosine Kinase Inhibitors in Chronic Myeloid Leukemia. Curr Hematol Malig Rep. 2016; 11(2):71–9. <u>https://doi.org/10.1007/</u> s11899-016-0309-2 PMID: 26922746
- Heiblig M, Rea D, Chretien ML, Charbonnier A, Rousselot P, Coiteux V, et al. Ponatinib evaluation and safety in real-life chronic myelogenous leukemia patients failing more than two tyrosine kinase inhibitors: the PEARL observational study. Exp Hematol. 2018; 67:41–8. https://doi.org/10.1016/j.exphem. 2018.08.006 PMID: 30195076
- **37.** Tao Z, Li SX, Shen K, Zhao Y, Zeng H, Ma X. Safety and Efficacy Profile of Neratinib: A Systematic Review and Meta-Analysis of 23 Prospective Clinical Trials. Clin Drug Investig. 2019; 39(1):27–43. https://doi.org/10.1007/s40261-018-0719-0 PMID: 30370488
- Paydas S. Management of adverse effects/toxicity of ibrutinib. Crit Rev Oncol Hematol. 2019; 136:56– 63. https://doi.org/10.1016/j.critrevonc.2019.02.001 PMID: 30878129
- Irvine E, Williams C. Treatment-, patient-, and disease-related factors and the emergence of adverse events with tyrosine kinase inhibitors for the treatment of chronic myeloid leukemia. Pharmacotherapy. 2013; 33(8):868–81. https://doi.org/10.1002/phar.1266 PMID: 23553655
- Liao BC, Lin CC, Lee JH, Yang JC. Update on recent preclinical and clinical studies of T790M mutantspecific irreversible epidermal growth factor receptor tyrosine kinase inhibitors. J Biomed Sci. 2016; 23 (1):86. https://doi.org/10.1186/s12929-016-0305-9 PMID: 27912760
- Noh YS, Yoon S, Kim SR, Lee KT. Jang IJ. A safety, pharmacokinetic, pharmacogenomic and population pharmacokinetic analysis of the third-generation EGFR TKI, olmutinib(HM61713), after single oral administration in healthy volunteers. Basic Clin Pharmacol Toxicol. 2019; 125(4):370–81. <u>https://doi.org/10.1111/bcpt.13262</u> PMID: 31125491
- 42. Smith PF, Krishnarajah J, Nunn PA, Hill RJ, Karr D, Tam D, et al. A phase I trial of PRN1008, a novel reversible covalent inhibitor of Bruton's tyrosine kinase, in healthy volunteers. Br J Clin Pharmacol. 2017; 83(11):2367–76. https://doi.org/10.1111/bcp.13351 PMID: 28636208
- Lee SK, Xing J, Catlett IM, Adamczyk R, Griffies A, Liu A, et al. Safety, pharmacokinetics, and pharmacodynamics of BMS-986142, a novel reversible BTK inhibitor, in healthy participants. Eur J Clin Pharmacol. 2017; 73(6):689–98. https://doi.org/10.1007/s00228-017-2226-2 PMID: 28265691
- Herman AE, Chinn LW, Kotwal SG, Murray ER, Zhao R, Florero M, et al. Safety, Pharmacokinetics, and Pharmacodynamics in Healthy Volunteers Treated With GDC-0853, a Selective Reversible Bruton's Tyrosine Kinase Inhibitor. Clin Pharmacol Ther. 2018; 103(6):1020–8. https://doi.org/10.1002/cpt. 1056 PMID: 29484638
- 45. Montassier E, Gastinne T, Vangay P, Al-Ghalith GA, Bruley des Varannes S, Massart S, et al. Chemotherapy-driven dysbiosis in the intestinal microbiome. Aliment Pharmacol Ther. 2015; 42(5):515–28. https://doi.org/10.1111/apt.13302 PMID: 26147207
- 46. Pal SK, Li SM, Wu X, Qin H, Kortylewski M, Hsu J, et al. Stool Bacteriomic Profiling in Patients with Metastatic Renal Cell Carcinoma Receiving Vascular Endothelial Growth Factor-Tyrosine Kinase Inhibitors. Clin Cancer Res. 2015; 21(23):5286–93. <u>https://doi.org/10.1158/1078-0432.CCR-15-0724</u> PMID: 26152743